

## **Author's response to reviews**

**Title:** Association between pain, neuropsychiatric symptoms, and physical function in dementia: a systematic review and meta-analysis

### **Authors:**

Annelore H van Dalen-Kok ([a.h.van\\_dalen@lumc.nl](mailto:a.h.van_dalen@lumc.nl))

Marjoleine J.C Pieper ([m.j.c.pieper@lumc.nl](mailto:m.j.c.pieper@lumc.nl))

Margot W.M de Waal ([m.w.m.de\\_waal@lumc.nl](mailto:m.w.m.de_waal@lumc.nl))

Albert Lukas ([albert.lukas@malteser.org](mailto:albert.lukas@malteser.org))

Bettina S Husebo ([bettina.husebo@isf.uib.no](mailto:bettina.husebo@isf.uib.no))

Wilco P Achterberg ([w.p.achterberg@lumc.nl](mailto:w.p.achterberg@lumc.nl))

**Version:**2 **Date:**24 November 2014

### **Author's response to reviews:**

1 Line numbering was added to the manuscript and page breaks were removed.

2 A statement of approval of the ethics committee is not included due to the fact that this research article is a systematic review of the literature. Therefore, no ethical approval was necessary.